Author's response to reviews

Title: Decreasing cholesterol levels in the community - lifestyle change with statin?

Authors:

Jorma Savolainen (jorma.savolainen@uef.fi)
Hannu Kautiainen (Hannu.Kautiainen@medcare.fi)
Leo Niskanen (Leo.Niskanen@fimea.fi)
Pekka Mäntyselkä (pekka.Mantyselka@uef.fi)

Version: 2 Date: 22 September 2014

Author's response to reviews:

Dear Editor

1. We have added information about approval of the ethics committee: "This study was carried out in compliance with the Helsinki Declaration. Ethical permission for the study was granted by the ethics committee of the Kuopio University Hospital (number 129/2009)."

2. We have added the line number

3. We do not have any repository of open access data

4. We have added the list of abbreviations

Sincerely

Jorma Savolainen, MD

Corresponding author